Author's response to reviews

Title: TRPV1 channel inhibition contributes to the antinociceptive effects of Croton macrostachyus extract in mice

Authors:

Télesphore B Nguelefack (nguelefack@yahoo.fr)
Rafael C Dutra (rafaeldutra@gmail.com)
Ana F Paszcuk (ana-dear@yahoo.com.br)
Edineia L Andrade (edineialemos@ig.com.br)
Joao B Calixto (calixto@farma.ufsc.br)

Version: 2 Date: 27 December 2014

Author's response to reviews:

1- The line numbers have been included.
2- The name of the ethical committee that approved the study is also included and highlighted in red (Please see the manuscript at page 5).